Author's response to reviews

Title: An investigation of somatosensory profiles in work related upper limb disorders: A case-control study protocol. Niamh Moloney 1, Toby Hall 2 3, Catherine Doody 1 1 UCD School of Public Health, Physiotherapy and Population Science, University College Dublin, Belfield, Dublin 4, Ireland 2 School of Physiotherapy, Curtin University of Technology, G.P.O. Box U1987, Perth, WA 6845, Australia 3 Manual Concepts, P.O. Box 1236, Booragoon, WA 6954, Australia

Authors:

Niamh Moloney (n.moloney@yahoo.com)
Catherine Doody (c.doody@ucd.ie)
Toby Hall (hallm62@me.com)

Version: 3 Date: 12 January 2010

Author's response to reviews:

The revisions made on the January 4th manuscript have not been altered. Formatting changes have been made as requested.